Author's response to reviews

Title: A Novel Mutation in the Glycogen Synthase 2 Gene in a Child with Hypoglycemia Due to Glycogen Storage Disease Type 0

Authors:

Ana Priscila Soggia (pri_soggia@yahoo.com.br)
Maria Lúcia Correa-Giannella (malugia@lim25.fm.usp.br)
Maria Angela Henriques Fortes (malugia@lim25.fm.usp.br)
Ana Mercedes Cavaleiro Luna (malugia@lim25.fm.usp.br)
Maria Adelaide Albergaria Pereira (adelaideap@uol.com.br)

Version: 5 Date: 27 November 2009

Author's response to reviews: see over
To:
Scott Edmunds, PhD
Senior Scientific Editor
BMC-series Journals

Dear Professor Edmunds

We are resubmitting the manuscript entitled "A Novel Mutation in the Glycogen Synthase 2 Gene in a Child with Glycogen Storage Disease Type 0". All suggestions were considered and revised.

ANSWER TO REVIEWER COMMENTS

REFEREE 1

1. In Page 4, hepatomegalia # hepatomegaly
   As suggested, the mistake was corrected.

2. In page 5, Glycemia and lactemia were measured by immunoenzymatic method # The plasma glucose and blood lactate were measured by the enzymatic method.
   The suggestion was accepted and the text was modified.

3. In page 7, the first paragraph in the discussion of R246X, the reference should be cited.
   The reference was inserted.

Hoping to hear from you very soon, I remain

Truly yours.

Maria Lúcia Corrêa-Giannella, MD, PhD